Author’s response to reviews

Title: Transvenous dual-chamber pacemaker implantation in patients with persistent left superior vena cava

Authors:

Teng Li (gdmc99@163.com)
Qiong Xu (47139772@qq.com)
Hong-tao Liao (hometown1975@163.com)
Dimitrios Asvestas (dimasvestas@gmail.com)
Konstantinos Letsas (k.letsas@gmail.com)
Yifu Li (yeefy@163.com)

Version: 2 Date: 11 Apr 2019

Author’s response to reviews:

Dear Editor:

We are truly grateful to you for all critical comments. Based on these comments and suggestions, we have made careful modifications to the last manuscript.

1. Thanks. Written informed consent was obtained from all individual participants included in the study. All data was obtained retrospectively. This study was approved and supervised by the Ethics Committee of Fuwai Hospital Chinese Academy of Medical Science and Guangdong General Hospital.

2. Manuscript had been uploaded as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files also were clean versions. Figures (and additional files) also were uploaded as separate files.